Author's response to reviews

Title: A randomised controlled feasibility trial for an educational school-based mental health intervention: Study protocol

Authors:

Katharine E Chisholm (KEC928@bham.ac.uk)
Paul Patterson (Paul.Patterson@bsmhft.nhs.uk)
Carole Torgerson (TorgerCJ@adf.bham.ac.uk)
Erin Turner (erin.turner@bsmhft.nhs.uk)
Max Birchwood (birchwom@adf.bham.ac.uk)

Version: 2 Date: 10 November 2011

Author's response to reviews:

Dear Editor

The acknowledgements and funding section has been updated to include a standardised version of text acknowledging the funding source.

Yours faithfully

Katharine Chisholm